Author's response to reviews

Title: Blood pressure in primary school children in Uganda: a cross-sectional survey

Authors:

Farah Kidy (Fkidy@yahoo.com)
Diana Rutebarika (rutedian@yahoo.co.uk)
Swaib A Lule (swaiblule@gmail.com)
Moses Kizza (moses.kizza@mrcuganda.org)
Amos Odiit (amosodiit@hotmail.com)
Emily L Webb (emily.webb@lshtm.ac.uk)
Alison M Elliott (alison.tom@infocom.co.ug)

Version: 2 Date: 3 September 2014

Author's response to reviews:

Line numbers have been added to the manuscript as requested, apologies for this omission!